Author's response to reviews

Title: Understanding the discriminant factors that influence the adoption and use of clinical communities of practice: the ECOPIH case

Authors:

David Lacasta-Tintorer (lacasta.david@gmail.com)
Souhel Flayeh Beneyto (souhelflayeh@gmail.com)
Josep M Manresa (imanresa@idiapigol.info)
Pere Torán-Monserrat (ptoran.bnm.ics@gencat.cat)
Ana I Jiménez-Zarco (ajimenezz@uoc.edu)
Joan Torrent-Sellens (jtorrent@uoc.edu)
Francesc Saigí-Rubió (fsaigi@uoc.edu)

Version: 2 Date: 30 May 2014

Author's response to reviews:

Dear Dr. Armee Valencia

Journal Editorial Office

Biomed Central

I am writing to you regarding the article “Understanding the discriminant factors that influence the adoption and use of clinical communities of practice: the ECOPIH case”. ID Manuscrito 1594689261310825.

Firstly, and on behalf of my co-authors, I would like to sincerely and humbly apologise for the inconvenience caused. Given the Editorial comments, we have made the following modifications:

1. We have specified the individual contributions of each author to the manuscript following the required format.

2. A new subsection called “Ethic” has been introduced at the end of Methods section.

3. Line numbering and page numbers have been included in our manuscript.

4. We have ensured that our manuscript follows the correct structure for this journal and article type.

Finally, we remain at your disposal for any further information you may need.

Yours sincerely,

The authors